Author's response to reviews

Title: Dental health behavior of parents of children using non-fluoride toothpaste: a cross-sectional study

Authors:

Junko Ota (ota-junko@mbp.nifty.com)
Tatsuo Yamamoto (yamamoto.tatsuo@kdu.ac.jp)
Yuichi Ando (ando@nih.go.jp)
Jun Aida (aidajun@m.tohoku.ac.jp)
Yukio Hirata (hirata@kdu.ac.jp)
Seishiro Arai (yama.tatsu.who@gmail.com)

Version: 3 Date: 19 November 2013

Author's response to reviews:

Dear Dr. Christopher Foote:

Please find enclosed a copy of our revised manuscript entitled, “Dental health behavior of parents of children using non-fluoride toothpaste: a cross-sectional study” which we would like to resubmit for publication in BMC Oral Health.

We wish to thank the reviewers for their careful critique of our original manuscript. We have made every effort to implement their recommendations, and we feel that the paper has been improved as a result. The revised parts were highlighted with blue. We hope you will find the manuscript suitable for publication in your journal.

Sincerely,

Tatsuo Yamamoto
Department of Dental Sociology,
Kanagawa Dental University Graduate School of Dentistry,
82 Inaoka-cho, Yokosuka,
Kanagawa 238-8580, Japan
TEL & FAX: +81 46 822 8838
e-mail: yamamoto.tatsuo@kdu.ac.jp